**Author’s response to reviews**

**Title:** Fever with lymphadenopathy: Kikuchi Fujimoto Disease- a great masquerader: A case report

**Authors:**
Chamara Dalugama (chamaradalugama@yahoo.com)
Indika Gawarammana (indika@sactrc.org)

**Version:** 1  **Date:** 05 Nov 2017

**Author's response**

Reviewer #1: Greeting

Thanks for allowing me to review this article.

It needs some changes

1. More organized case presentation
2. Kikuchi-Fujimoto Disease (KFD) is not a rare disease.
3. Discussion must be more focused and compare with other cases

Regards

Thank you for reviewing my manuscript

I have made some changes in the case presentation

I changed the KFD is a rare disease to an uncommon disease

I made the discussion more focused and making comparisons with other cases

Reviewer #2: This paper is a case report that focuses on Kikuchi Fujimoto Disease. Although the topic is interesting; the presentation of the case was poor. The pathological diagnosis is important in Kikuchi Fujimoto Disease, however there is no pathological image in this case report.

I have added a pathological slide of the lymph node specimen